Author's response to reviews

Title: Transformation of human bronchial epithelial cells alters responsiveness to inflammatory cytokines

Authors:

- Gregory M Loewen (gregory.loewen@roswellpark.org)
- Erin Tracy (erin.tracy@roswellpark.org)
- Frederic Blanchard (frederic.blanchard@univ-nantes.fr)
- Dongfeng Tan (dongfeng.tan@uth.tmc.edu)
- Jihnhee Yu (yuj@roswellpark.org)
- Sameera Raza (straza@buffalo.edu)
- Sei-Ichi Matsui (selichi.matsui@roswellpark.org)
- Heinz Baumann (heinz.baumann@roswellpark.org)

Version: 3 Date: 27 October 2005

Author's response to reviews:

Thursday, October 27, 2005

BMC-Cancer

Re: MS: 6110777437520870 - Transformation of human bronchial epithelial cells alters responsiveness to inflammatory cytokines

Dear Editor,

My colleagues and I are pleased to learn that the manuscript is acceptable in principle. The reviewer has recommended acceptance as is. The editorial suggestions for format changes have been made to the manuscript and to the text in the submission system.

We would like to thank you and the reviewer, Dr. Frank Buhling, for your consideration

Sincerely yours

Heinz Baumann
Department of Molecular and Cellular Biology
Roswell Park Cancer Institute
Elm & Carlton Streets
Tel. (716) 845 4587
Fax (716) 845 5908
Email: heinz.baumann@roswellpark.org